
Mauna Loa Solar Observatory Observer's Log

Sun May 3 17:07:39 GMT 1998

Year: 98 Doy: 123 Observer: koon

WEATHER COMMENT: Sun May 3 17:07:47 GMT 1998

Scattered thin cirrus and thin cirrostratus, wind=5 mph from the South,

temp=48 F.

 Sun May
 3 17:13:00 GMT 1998
 MKIII
 Start Patrol

 Sun May
 3 17:13:11 GMT 1998
 CHIP
 CHIP Start Patrol

 Sun May
 3 17:13:11 GMT 1998
 CHIP
 CHIP Start Patrol

LOW-L COMMENT: Sun May 3 17:27:03 GMT 1998

Found L00511 popped out of temp drive #1 as it should be. But the program crashed instead of switching over, the screen was gray and the instrument was stowed. L00512 was still inside temp drive #0 but was apparently unused. I removed the tapes from the drives, turned off power to the computer and the drives, switched the cables and the terminator on the drives so that temp drives are setup exactly opposite from their normal configuration. So now if temp drive #0 still gives problems several times it is very likely that there is a problem with that drive rather than the interfaces or software. Previously I had only switched the switch assignments on the backs of the drives but now the complete "switch" is in place. L00512 is taking data in temp drive #0, L00513 is installed in temp drive #1.

Sun May 3 18:00:24 GMT 1998 PICS Flat Sun May 3 18:01:58 GMT 1998 End Flat PICS Sun May 3 18:02:03 GMT 1998 Bias CHIP Sun May 3 18:03:04 GMT 1998 End Bias CHIP Sun May 3 18:03:19 GMT 1998 Water CHIP Sun May 3 18:04:08 GMT 1998 CHIP End Water Sun May 3 18:53:10 GMT 1998 MKIII Start Cal Sun May 3 19:01:19 GMT 1998 PICS Flat Sun May 3 19:02:11 GMT 1998 CHIP Bias Sun May 3 19:02:45 GMT 1998 PICS End Flat Sun May 3 19:03:22 GMT 1998 End Bias CHIP Sun May 3 19:03:32 GMT 1998 CHIP Water Sun May 3 19:04:15 GMT 1998 End Water

PSPT COMMENT: Sun May 3 19:31:14 GMT 1998

Appeared to start up OK and take some data, but now it appears to be waiting for better sky conditions according to the screen messages that keep

scrolling by (guider not locked on the sun). WEATHER COMMENT: Sun May 3 19:33:14 GMT 1998

Cirrostratus is getting thicker.

PICS COMMENT: Sun May 3 19:52:49 GMT 1998

Pausing due to clouds.

MKIII COMMENT: Sun May 3 19:52:59 GMT 1998

Stopping due to clouds.

1998 PICS End Patrol Sun May 3 20:02:05 GMT 1998 CHIP Gain Sun May 3 20:07:55 GMT 1998 CHIP End Gain Sun May 3 20:08:05 GMT 1998 Bias CHIP Sun May 3 20:08:59 GMT 1998 End Bias CHIP Sun May 3 20:09:08 GMT 1998 CHIP Water Sun May 3 20:09:48 GMT 1998 CHIP End Water COMMENT: Sun May 3 20:21:31 GMT 1998 Extended the dome slot. PSPT COMMENT: Sun May 3 20:21:41 GMT 1998 I was going to stop the program due to clouds, but when I was able to get a "g" to go through the program responded by appearing to try to continue instead of guitting like it was supposed to do. PSPT COMMENT: Sun May 3 20:48:03 GMT 1998 I'm not sure if any data were taken since my last message, but the guider unlocked messages started coming back again. This time I type q then hit Return and the program ended right away with the psp> prompt. So I used Obs to exit observations. CHIP COMMENT: Sun May 3 20:52:50 GMT 1998 Thick cloud mass moving over, pausing. Sun May 3 20:55:34 GMT 1998 CHIP CHIP End Patrol PICS COMMENT: Sun May 3 21:23:07 GMT 1998 We have been seeing what appeared to be vignetting noticeable on the West sides of the PICS and occulted images for a month or so. I found that the new O2A is indeed vignetting the light as evidenced by a spillover of light off the edge of the lens towards the spar Pier. The M2 mirror before that shows the light to be slightly off center in the same direction but there is no vignetting there. The first mirror after the O2A shows light spilling over on the same side as the O2A, this is the new mirror that directs light out of the spar. I haven't noticed any vignetting earlier in the day. I'll try to follow the direction of the lightpath cross-section throughout a normal day, probably next Wednesday. Sun May 3 22:00:02 GMT 1998 CHIP ending tape COMMENT: Sun May 3 22:03:49 GMT 1998 TAPES: ***** MKIII: H01818 PICS: P01184 CHIP: C00574 LOWL: L00512 in temp drive #0 Sun May 3 22:04:59 GMT 1998 MkIII 17 13.rawmk3 17 45.rawmk3 18 17.rawmk3 18 49.rawmk3 19 32.rawmk3

18 20.rawmk3

18_23.rawmk3

18 56.rawmk3 19 35.rawmk3

19 38.rawmk3

19 02.rawmk3

17 16.rawmk3

17 19.rawmk3

17 48.rawmk3

17 51.rawmk3

17_22.rawmk3	17_54.rawmk3	18_26.rawmk3	19_08.rawmk3	19_40.rawmk3
17_24.rawmk3	17_57.rawmk3	18_29.rawmk3	19_11.rawmk3	19_43.rawmk3
17_27.rawmk3	18_00.rawmk3	18_32.rawmk3	19_14.rawmk3	19_46.rawmk3
17_30.rawmk3	18_02.rawmk3	18_35.rawmk3	19_17.rawmk3	19_49.rawmk3
17_33.rawmk3	18_05.rawmk3	18_38.rawmk3	19_20.rawmk3	c18_53.rawmk3
17_36.rawmk3	18_08.rawmk3	18_40.rawmk3	19_23.rawmk3	c18_59.rawmk3
17_39.rawmk3	18_11.rawmk3	18_43.rawmk3	19_26.rawmk3	c19_05.rawmk3
17_42.rawmk3	18_14.rawmk3	18_46.rawmk3	19_29.rawmk3	